Title: A case of spontaneous regression of bone metastasis from renal cell carcinoma; Case Report

Authors:

Takahiro Nakajima (nakajii@fc.med.miyazaki-u.ac.jp)
Makoto Suzuki (smakoto@chiba-cc.jp)
Soichiro Ando (sando@chiba-cc.jp)
Tomohiko Iida (tiida@chiba-cc.jp)
Akinobu Araki (aaraki@chiba-cc.jp)
Takehiko Fujisawa (fujisawat@faculty.chiba-u.jp)
Hideki Kimura (lung@chiba-cc.jp)

Version: 5 Date: 28 November 2005

Author's response to reviews: see over
The BioMed Central Editorial Team,
Tel: +44 (0)20 7631 9921; Fax: +44 (0)20 7631 9923

Dear Dr. Iratxe Puebla, Senior Assistant Editor, BMC Cancer

Enclosed please find a revised manuscript by Takahiro Nakajima, et al #4495155486528136 entitled: “Spontaneous regression of bone metastasis from renal cell carcinoma: A case report”. We made changes in the text thoroughly according to referee’s comments. Now we believe it suitable for publication for “BMC Cancer”. We are looking forward to your reply.

Thank you for your consideration in this matter.

Yours sincerely

Takahiro Nakajima, MD

I would like to explain below how we have responded to your comment.

1. I have changed the title to Spontaneous regression of bone metastasis from renal cell carcinoma: A case report”.
2. I have changed the structure of the abstract into Background, Case Presentation, and Conclusions.
3. I also have changed the structure of the manuscript section include Case Presentation, Competing interests, Authors’ contributions.
4. I have changed the reference section using the numbers in square brackets.